
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Mon Apr 29 16:37:16 GMT 2013
Year: 13  Doy: 119
Observer: berkey
Mon Apr 29 16:39:01 GMT 2013: PSPT Start Patrol
Mon Apr 29 16:40:14 GMT 2013    CHIP     Start Patrol
Mon Apr 29 16:42:02 GMT 2013 COMP Start Patrol
WEATHER COMMENT: Mon Apr 29 16:41:13 GMT 2013
Clear skies, temp 43f, wind 8mph from SW
___end___
Mon Apr 29 16:52:09 GMT 2013    MKIV    Start Patrol
Mon Apr 29 17:14:58 GMT 2013 COMP End Patrol
Mon Apr 29 17:16:30 GMT 2013    MKIV    End Patrol
Mon Apr 29 17:21:47 GMT 2013    MKIV    Start Patrol
Mon Apr 29 17:22:52 GMT 2013 COMP Start Patrol
NOTE BY BEN: Mon Apr 29 17:49:20 GMT 2013
U12 is acting up this morning, I cant get it to stay happy.
___end___
Mon Apr 29 17:51:40 GMT 2013    MKIV    End Patrol
Mon Apr 29 18:01:35 GMT 2013    CHIP     LSD
Mon Apr 29 18:03:24 GMT 2013    CHIP     End LSD
Mon Apr 29 18:03:32 GMT 2013    CHIP     BiasLSD
Mon Apr 29 18:04:11 GMT 2013    CHIP     End BiasLSD
Mon Apr 29 18:04:20 GMT 2013    CHIP     Bias
Mon Apr 29 18:05:10 GMT 2013    CHIP     End Bias
Mon Apr 29 18:05:20 GMT 2013    CHIP     ReStart Patrol
NOTE BY BEN: Mon Apr 29 18:06:49 GMT 2013
Ideling comp and mk4 for cirrus.
___end___
Mon Apr 29 18:08:01 GMT 2013 COMP End Patrol
Mon Apr 29 20:29:33 GMT 2013    CHIP     End Patrol
NOTE BY BEN: Mon Apr 29 20:35:29 GMT 2013
Still suffering U12 and cirrus issues
___end___
Mon Apr 29 20:36:06 GMT 2013    CHIP     Start Patrol
Mon Apr 29 20:48:32 GMT 2013Mon Apr 29 20:48:36 GMT 2013        CHIP     End Patrol
NOTE BY BEN: Mon Apr 29 20:48:18 GMT 2013
Idleing chip and pspt for orragraphic clouds
___end___
: PSPT Abort Patrol
Mon Apr 29 20:59:34 GMT 2013    CHIP     Start Patrol
Mon Apr 29 21:02:18 GMT 2013: PSPT Start Patrol
Mon Apr 29 21:11:37 GMT 2013: PSPT Abort Patrol
Mon Apr 29 21:20:35 GMT 2013    CHIP     End Patrol
CoMP COMMENT BY BEN: Tue Apr 30 00:39:58 GMT 2013
CoMP occulter is now #27, changes have been made to the VI to remember this



___end___
GONG COMMENT BY BEN: Tue Apr 30 02:09:51 GMT 2013
Replaced the diaphram in the gong clean air pump.  It was starting to fail
I also did the normal weekly pm checks
___end___
Tue Apr 30 02:42:15 GMT 2013 
        MkIV    
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